Author's response to reviews

Title: The perception of primary health care professionals about education in diabetes melitus, Brazil

Authors:

Heloisa C Torres (heloisa@enf.ufmg.br)
Marta A Amaral (marta@enf.ufmg.br)
Brani R Rosenberg (brani@emsp.fiocruz.br)
Regina C Bodstein (bodstein@ensp.fiocruz.br)

Version: 4 Date: 23 February 2010

Author's response to reviews: see over
Belo Horizonte, February 23, 2010

Dear Editor of the Journal,

You have to inform that all authors contributed to the work and agree with the article be submitted to the journal BMC Public Health. Place in the provision for any clarification. We are awaiting news. We emphasize that this work was not published in another journal. There is no conflict of interest.

Cordially,

Profa. Dra. Heloisa de Carvalho Torres
Departamento de Enfermagem Aplicada da Escola de Enfermagem-UFMG